Author’s response to reviews

Title: Social and environmental conditions related to Mycobacterium leprae infection in children and adolescents from three leprosy endemic regions of Colombia

Authors:
Héctor Serrano-Coll (serrano.hector@uces.edu.co)
Hugo Mora (hugorenemora@gmail.com)
Juan Beltrán (jbeltran@ces.edu.co)
Malcolm Duthie (Malcolm.Duthie@idri.org)
Nora Cardona-Castro (ncardona@ces.edu.co)

Version: 2 Date: 18 Feb 2019

Author’s response to reviews:
Dear Dr. Kevin Escandón-Vargas, MD MSc
Associate Editor - BMC Infectious Diseases

https://bmcinfectdis.biomedcentral.com/

We appreciate all your comments and help with this submission. We added the Figure 1 as you asked.

Best wishes,

Nora Cardona-Castro